Author’s response to reviews

Title: Liver transplantation in Jehovah’s witnesses: 13 consecutive cases at a single Institution

Authors:

Diego Costanzo (zgundam.dc@gmail.com)
Maria Bindi (l.bindi@ao-pisa.toscana.it)
Davide Ghinolfi (dghinolgi@yahoo.com)
Massimo Esposito (m.esposito@ao-pisa.toscana.it)
Francesco Corradi (f.corradi@med.unipi.it)
Francesco Forfori (f.forfori@unipi.it)
Paolo De Simone (p.desimone@unipi.it)
Andrea De Gasperi (andrea.degasperi@mpcnet.it)
Gianni Biancofiore (g.biancofiore@med.unipi.it)

Version: 6 Date: 13 Jan 2020

Author’s response to reviews:

all the required changes were made.

we included a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript and the presented copy of the manuscript is clean